Author's response to reviews

Title: The Healthy Lifestyle and Personal Control Questionnaire (HLPCQ): A novel tool for assessing the efficacy of stress management and health promotion interventions.

Authors:

Christina Darviri (cdarviri@yahoo.com)
Evangelos C Alexopoulos (ecalexop@med.uoa.gr)
Artemios K Artemiadis (artemiad@med.uoa.gr)
Xanthi Tigani (xtigani@med.uoa.gr)
Christina Kraniotou (chr1s_kran@hotmail.com)
Panagiota Darvyri (pandarviri@yahoo.gr)
George P Chrousos (chrousge@med.uoa.gr)

Version: 2 Date: 11 June 2014

Author's response to reviews: see over
Dear Editor,

As a response to your email we made the necessary corrections.

1) We included line and page numbers.
2) We included an “Acknowledgements” and “Funding” section.
3) We included a statement detailing the ethical approval that has been obtained.

Thank you in advance,
Yours sincerely
Pr. Darviri Christina